

GOPEN ACCESS

Citation: Jang H, Polgreen PM, Segre AM, Pemmaraju SV (2021) COVID-19 modeling and non-pharmaceutical interventions in an outpatient dialysis unit. PLoS Comput Biol 17(7): e1009177. https://doi.org/10.1371/journal.pcbi.1009177

Editor: Roger Dimitri Kouyos, University of Zurich, SWITZERLAND

Received: August 16, 2020

Accepted: June 14, 2021

Published: July 8, 2021

Copyright: © 2021 Jang et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All data files are available from the Kaggle database (URL: <u>https://</u> www.kaggle.com/hankyujang/healthcarepersonnel-movement-data, DOI:10.34740/kaggle/ dsv/1397235).

Funding: Funding for this research was provided as part of CDC the MInD Healthcare group under cooperative agreement U01CK000531 and associated Covid19 supplemental funding. Authors PMP, AMS, SVP were awarded the grant and all 4 authors were supported by the grant. URL: https:// www.cdc.gov/hai/research/MIND-Healthcare.html **RESEARCH ARTICLE**

COVID-19 modeling and non-pharmaceutical interventions in an outpatient dialysis unit

Hankyu Jang¹, Philip M. Polgreen², Alberto M. Segre¹, Sriram V. Pemmaraju¹*

 Department of Computer Science, The University of Iowa, Iowa City, Iowa, United States of America,
 Department of Internal Medicine, Division of Infectious Diseases, Carver College of Medicine, The University of Iowa, Iowa City, Iowa, United States of America

* sriram-pemmaraju@uiowa.edu

Abstract

This paper describes a data-driven simulation study that explores the relative impact of several low-cost and practical non-pharmaceutical interventions on the spread of COVID-19 in an outpatient hospital dialysis unit. The interventions considered include: (i) voluntary selfisolation of healthcare personnel (HCPs) with symptoms; (ii) a program of active syndromic surveillance and compulsory isolation of HCPs; (iii) the use of masks or respirators by patients and HCPs; (iv) improved social distancing among HCPs; (v) increased physical separation of dialysis stations; and (vi) patient isolation combined with preemptive isolation of exposed HCPs. Our simulations show that under conditions that existed prior to the COVID-19 outbreak, extremely high rates of COVID-19 infection can result in a dialysis unit. In simulations under worst-case modeling assumptions, a combination of relatively inexpensive interventions such as requiring surgical masks for everyone, encouraging social distancing between healthcare professionals (HCPs), slightly increasing the physical distance between dialysis stations, and—once the first symptomatic patient is detected—isolating that patient, replacing the HCP having had the most exposure to that patient, and relatively short-term use of N95 respirators by other HCPs can lead to a substantial reduction in both the attack rate and the likelihood of any spread beyond patient zero. For example, in a scenario with $R_0 = 3.0, 60\%$ presymptomatic viral shedding, and a dialysis patient being the infection source, the attack rate falls from 87.8% at baseline to 34.6% with this intervention bundle. Furthermore, the likelihood of having no additional infections increases from 6.2% at baseline to 32.4% with this intervention bundle.

Author summary

As we write this, the COVID-19 pandemic has essentially taken over the world, with more than 20 million cases spread over 216 countries. A big concern for policy makers all across the world has been the impact of COVID-19 on healthcare systems and whether these systems can cope with the enormous strain placed on them by COVID-19. In this paper, we consider the spread of COVID-19 in a specific healthcare setting: the outpatient dialysis unit. Hemodialysis patients are extremely vulnerable to infections in large part due to

The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

multiple immune-system deficiencies associated with renal failure and hemodialysis. Hemodialysis facilities also increase the risk of COVID-19 transmission because each patient is in frequent, close contact with other patients and healthcare personnel. Thus, a dialysis unit can be seen as a microcosm for the worst-case impacts of COVID-19 in a healthcare setting. In this manuscript, we show via high-fidelity modeling and simulations that under pessimistic modeling assumptions, there is a combination of relatively simple, inexpensive, and practical non-pharmaceutical interventions that can substantially lower the impact of COVID-19 in the dialysis unit. Our simulations are based on fine-grained healthcare personnel movement data that we make available for other modelers to use.

Introduction

As of August 13, 2020, 5.2M people in the United States were infected with the COVID-19 virus, with more than 166K of these cases resulting in death of the patient [1]. Using data from 13 US states, the CDC estimates that among adults hospitalized due to COVID-19 during March 1–May 31, 2020, 5.9% were HCPs and of these, nursing-related occupations (36.3%) represented the largest proportion [2] The CDC reports that 132K US healthcare personnel (HCPs) were infected as of August 13, though this is likely a substantial underestimate because information on whether an individual was an HCP was available for only 22.2% of the total cases. Thus healthcare settings represent a significant locus of COVID-19 transmission for HCPs.

Infections are the second leading cause of mortality among hemodialysis patients [3], and rates of sepsis are approximately 100 times greater than in the general population [4]. Almost 20% of patients on hemodialysis develop pneumonia in their first year [5], and rates of pneumonia among dialysis patients are three to five times higher than in the general population [6, 7]. Risk of infections and adverse outcomes attributable to infections are in large part due to multiple immune-system deficiencies associated with renal failure and hemodialysis [8–10].

Patients undergoing hemodialysis are especially at risk for acquiring COVID-19, and to date, several COVID-19 outbreaks have occurred in dialysis centers in multiple countries [11–15]. Hemodialysis facilities may increase the risk of COVID-19 transmission and other respiratory droplet infections because each patient is in frequent, close contact with other patients and HCPs: most dialysis centers are large spaces without separation between patients [16]. The efficiency of transmission in this setting is also increased by the fact that a small team of HCPs usually attends to a larger set of patients, and that dialysis unit HCPs are difficult to replace due to their specialized skills. Furthermore, patients undergoing dialysis experience additional healthcare exposures beyond their usual hemodialysis center. For example, patients on dialysis are frequently hospitalized and often reside in nursing homes [3, 17, 18]. In addition to acquiring COVID-19, dialysis patients may be at greater risk of developing more severe cases as they often suffer from many chronic diseases, e.g., diabetes and hypertension, that have been linked to greater COVID-19 morbidity and mortality [19].

This paper explores the effectiveness of simple, practical, and low-cost non-pharmaceutical interventions (NPIs) in reducing the spread of COVID-19 in one specific healthcare setting: the outpatient dialysis unit. Our work uses computational modeling and discrete-event simulations to evaluate the effectiveness of six specific simple, low-cost NPIs in reducing the spread of COVID-19 in a dialysis unit. Our high-fidelity simulations are based on fine-grained HCP movement and interaction data collected in an outpatient hospital dialysis unit at the University of Iowa Hospitals and Clinics (UIHC). We have used this data in previous work, wherein

we have simulated the spread of Methicillin-resistant *Staphylococcus aureus* (MRSA) in a dialysis unit and investigated the effectiveness of interventions such as architectural changes to the unit [20]. In this work, we test the candidate NPIs both alone and in combination, using a set of disease models that represent alternate possibilities of how COVID-19 is transmitted and unfolds. The NPIs under study are: (*i*) voluntary self-isolation; (*ii*) active syndromic surveillance and compulsory isolation; (*iii*) the use of masks (HCPs, patients) and/or N95 respirators (HCPs); (*iv*) improved social distancing among HCPs; (*v*) increased physical separation between dialysis stations; and (*vi*) patient isolation combined with the preemptive isolation of exposed HCPs. A description of these NPIs appears in the section Methods: NPIs.

As we evaluate these NPIs, we focus not just on their effect on COVID-19 spread, but also on the cost of implementing these NPIs. For the NPIs we consider, cost can be measured by the number or HCPs temporarily replaced and the duration of this replacement, the number of N95 respirators used and the duration of this usage, and the number of dialysis patients sent to isolation unit. The main takeaway from our work is that even though under pre-COVID-19 usage conditions COVID-19 can have near-100% attack rate in the dialysis unit within 30 days of the first agent becoming infected, it is possible to design and implement a low-cost NPI bundle that leads to a substantial reduction in the attack rate and likelihood of COVID-19 spread in the dialysis unit.

Methods

Ethics statement

We did not gather any patient data from our instrumentation of the dialysis unit. All HCP movement that we gathered was completely anonymous. Badges were arbitrarily distributed to HCPs at the start of each day, ensuring that there was no necessary connection between HCPs with the same badge ID across different days. For all these reasons, this project was human subjects research exempt.

The UIHC dialysis unit serves patients six days a week. Patients typically dialyze three times a week either Monday, Wednesday, and Friday (MWF) or Tuesday, Thursday, and Saturday (TThS)(either MWF or TThS) in one of three daily blocks (morning, afternoon, or evening). Interactions between patients and HCPs form an implicit underlying *contact network* which, along with an explicit model of how the infection is transmitted between agents, form the basis for agent-based simulation. A typical simulation study, then, introduces a single infected agent into a virtual dialysis unit defined by the contact network, the disease model, and the particular set of NPIs being tested. Outcomes from each trial, typically in the form of total number of infected individuals and the cost of the NPI, can be aggregated over multiple replicates having the same experimental conditions and then compared across conditions to determine the effectiveness of the NPIs under study.

Temporal contact networks

As described in [20], ten days of deidentified HCP movement and interaction data were gathered at a dialysis facility at the UIHC in the fall of 2013 [21]. Twenty two line powered *beacons* were placed in static locations including at each dialysis station, and rechargeable *badges* were distributed to HCPs in the unit. There are 9 dialysis chairs placed along the perimeter of a rectangular room, a nurses' station in the center, a bathroom and two handwashing stations. For a schematic of the dialysis unit, including locations of the beacons, consult our previous work [20] (Fig 1). Each badge broadcasts its identity every few seconds, while beacons record and timestamp each message received, along with an indication of its received signal strength index (RSSI). RSSI increases with proximity, a fact later used to reconstruct the location and





movement of HCPs within the unit. Because patients were not badged, dialysis sessions were imputed based on knowledge of the unit's usual patient schedule and the observation that a HCP typically spends a prolonged time at a dialysis chair at the start and end of each dialysis session [20, 22]. We cast this problem as a binary classification problem whose aim is to classify, for each chair, every time window as either the start/end of a dialysis session in that chair or not. This classification is based on the distance profiles of all HCPs to with respect to a chair during a time window. To implement this idea, we precompute the distances of all HCPs to all the dialysis chairs over the observed timesteps. Then, for each chair, we manually select HCP distance profiles that correspond to long-duration, close contacts and mark these as positive instances. We also manually select distance profiles that are not long-duration close contacts and mark these as negative instances. We then train a binary classifier on these positive and negative instances. Finally, for each dialysis chair and every time window, we apply the trained classifier to classify the distance profiles of HCPs to that chair during that time window. This process identifies, for every dialysis chair some time windows as the start/end times of patient dialysis sessions. This process is described in more detail in [20]. Moreover, because data were collected anonymously thereby precluding linkage across multiple days, our simulations rely on repeating a single day's data to create a 30 day simulation.

Our system for imputing dialysis sessions detected between 13 and 20 patient sessions over 6 deployment days [20]. To simulate a worst-case scenario, we picked the day with 20 dialysis sessions to repeat. Thus 20 patients had dialysis sessions on a MWF schedule and another 20 on a TThS schedule. There were 11 HCPs present in the data on this day. We generated 30 days of simulated unit operations by randomly assigning patients to dialysis stations on a consistent (morning, afternoon, or evening) block schedule. This results in 6–7 dialyzing patients per block, with a total population of 51 agents (40 patients and 11 HCPs) in the simulation. By replaying the HCP trajectories in parallel with the imputed patient dialysis sessions, we reconstructed all contacts (where a *contact* is defined as two agents, either HCPs and/or patients, within 6 feet of each other) between pairs of agents at 8 second intervals.

From these contact data, we generated undirected, unweighted, temporal contact networks $G^{d,t} = (V^{d,t}, E^{d,t})$ where $d, 0 \le d \le 29$, represents a day in simulation, and $t, 0 \le t \le 6821$, represents each 8 second timestep within a roughly 15 hour day starting at 5:31 AM. Here $V^{d,t}$ denotes the set of nodes (patients and HCPs) that participate in a contact on day d, at timestep t and $E^{d,t}$ consists of all edges $\{x, y\}$, representing pairwise contacts between agent x and agent y occurring on day d, at timestep t. Note that the graphs $G^{d,t}$ and $G^{d',t}$, for $d \ne d'$, are quite similar because agent-agent contacts are obtained by replaying identical HCP movement patterns every day. However, there are differences between these graphs that are induced by differences in start/end times of dialysis sessions and assignment of patients to dialysis chairs. Let $V = \bigcup_{d,t} V^{d,t}$ be the collection of patients and HCPs over the course of the entire simulation. Let V_H and

 $V_P = V \setminus V_H$ denote the set of HCP nodes and patient nodes, respectively; $|V_H| = 11$ and $|V_P| = 40$. Let $E = \bigcup_{d,t} E^{d,t}$ denote the set of contacts throughout the simulation period. To each edge $\{x, y\}$ in *E* we assign a positive integer weight that is the number of times the contact between agent *x* and agent *y* occurred over all the temporal graphs $G^{d,t}$. Finally, let G = (V, E) denote the static, undirected, weighted, contact network of all HCPs and patients.

Shedding models

Fig 1 shows a probabilistic state transition diagram of an individual agent's disease states. At each timestep of the simulation, a susceptible agent (in state *S*) will either become infected or remain susceptible, depending on their interactions with other agents and in accordance with the shedding model (see below). Probabilistically, 60% of the agents who become infected do so with symptoms and transition into the presymptomatic state *P* [23–27]. The remaining 40% become asymptomatically infected and transition into state *A*. The time, *period*_P, that an agent spends in state *P* is geometrically distributed with an expected length of 6 days [23, 28–31]. Following state *P*, an agent spends *period*_I days—geometrically distributed with expected length of 7 days—in the symptomatically infected state *I* [32–36]. Symptoms emerge in agents immediately after *period*_P days, i.e., as they transition from state *P* to state *I*. Agents who enter state *A* stay in that state for a total of *period*_P + *period*_I days. In other words, we assume that there is no difference in the total period of infection between patients who are symptomatic and asymptomatic patients. On leaving state *I* or *A*, agents recover and become immune to reinfection. This is represented by state *R*.

In this study, we consider a viral shedding model with an exponential ramp up followed by an exponential ramp down in daily shedding level (see Fig 2). Specifically, we use the notation exp/exp (f%) to denote a viral shedding model with the following properties:

- (i). The shedding level for every COVID-19 cases, whether symptomatic or not, exponentially ramps up daily, reaches a maximum, and then exponentially ramps down daily.
- (ii). The length of the ramp up period is assumed to be exactly $period_P$ and the length of the ramp down period is assumed to be exactly $period_I$. This implies that for a symptomatic



Viral shedding models

Fig 2. Viral shedding models. The bar charts show the shedding level of an infected individual whose presymptomatic period and symptomatic period are 6 and 7 days, respectively. The percentage in the parenthesis of each model corresponds to the percentage of shedding during the presymptomatic state (*P*) relative to the total shedding volume. The peak shedding level differs in the two models, but the total shedding volume remains the same.

https://doi.org/10.1371/journal.pcbi.1009177.g002

COVID-19 case, the day of maximum viral shedding is the first day in which symptoms emerge.

(iii). The exponential ramp up and ramp down factors are chosen so that *f*% of the total shedding occurs in the presymptomatic state.

The ramp up and ramp down aspects of this model and the timing of the peak viral shedding are inspired by experimental evidence for similar behavior in influenza and SARS, where viral loads initially build, peak at onset of symptoms, and decline thereafter [36-38].

Different instantiations of this model are obtained by using different values of f. In this paper, we show results for f = 20 and f = 60 (see Fig 2). These values of f yield the percentage of transmission occurring prior to symptom onset in the 30%—70% range [23, 36, 39]. The proportion of presymptomatic viral shedding f is somewhat lower than the proportion of presymptomatic transmission because there remain fewer susceptible agents in the latter days of the simulation. In Section "Calibrating the shedding model", we describe our shedding model more precisely, explain how it is parameterized, and describe a method to solve for model parameters so that the model is appropriately calibrated.

We assume that the infectiousness of an individual who ends up in state *A* is 75% of the infectiousness of an individual who is symptomatically infected over the course of their illness. We assume that the 40% of individuals who end up in state *A* are uniformly 75% less infectious than those in state symptomatically infected over the course of their illness, reflecting what some have found to be reduced viral loads and a concomitant reduction in viral shedding on the part of asymptomatic agents [23–26, 38, 40, 41].

Calibrating the shedding model

Our shedding model needs to be calibrated so that three constraints are met. First, we require that the shedding during the presymptomatic period be a fraction f of the total shedding. Second, we require that the resulting simulation yields a given target R_0 in expectation in the scenario where the initial infected is a dialysis patient in the morning session of the first day. We use 2, 2.5, and 3 as target values for R_0 and this is in line with estimated R_0 values for COVID-19 [23, 31, 42, 43]. Third, for two alternate shedding models exp/exp (f_1 %) and exp/exp (f_2 %) with the same target R_0 , we also require that the total shedding volume for both models be identical. For example, for a particular target R_0 value, (e.g., $R_0 = 3$), we require that the exp/exp (20%) and exp/exp (60%) models both have the same total shedding volume, though this is achieved via different "shapes," as can be seen in Fig 2. This last requirement is to ensure that there are no subtle differences in attack rates across the two alternate models that are caused by differences in total shedding volume.

Let s_d denote the *shedding level* of an agent u on day d of that agent's infection. Without loss of generality, we assume that the maximum value of s_d is 1 unit. Let the shedding ramp up parameter be denoted $\beta > 1$ and the shedding ramp down parameter be $\gamma > 1$. This implies that the shedding levels are given by the sequence

$$\ldots \frac{1}{\beta^2}, \frac{1}{\beta}, 1, \frac{1}{\gamma}, \frac{1}{\gamma^2}, \ldots$$

For a given value of *f*, the parameters β and γ are related by the equation

$$\left(1 - \frac{f}{100}\right) \cdot \left(\sum_{i=1}^{6} \frac{1}{\beta^i}\right) = \frac{f}{100} \cdot \left(\sum_{j=0}^{6} \frac{1}{\gamma^j}\right). \tag{1}$$

As a result, for a given *f*, we can view s_d as being determined by a single parameter γ .

Now consider a contact between an infected agent u and a susceptible agent v on day d of u's infection. The probability p_d of u transmitting infection to v via this contact is proportional to u's shedding level s_d on that day, i.e.,

$$p_d = \alpha \cdot s_d \tag{2}$$

for a scaling parameter α . Since $s_d = 1$ on the day d at which the shedding level is maximum (which is also the day on which symptoms emerge), the parameter α has the interpretation that it is the probability of transmitting infection via a single contact on the day on which symptoms first emerge. Let $L = period_E + period_I - 1$ denote the index of the last day that an individual is shedding. Let $x_d(u, v)$ denote the number of contacts u has with v on day d. For each contact, the disease is transmitted with the probability of p_d . Therefore, the event that ufails to infect v on day d can be expressed as a conjunction of independent events in which ufails to infect v in each contact. Letting $f_d^{u,v}$ denote the probability that u fails to infect v on day d, we get:

$$f_d^{u,v} = (1 - p_d)^{x_d(u,v)} \tag{3}$$

Let $P_{u,v}$ denote the probability that *u* infects *v* at some point in the simulation; this can be computed by 1 minus the probability that *u* fails to infect *v* from day 0 to day *L*:

$$P_{u,v} = 1 - f_0^{u,v} \cdot f_1^{u,v} \cdots f_L^{u,v}$$
(4)

At this point, we have computed the probability of an infectious agent *u* infecting a susceptible agent *v*. Now referring to Fig.3, let $V'_p \subseteq V_p$ denote the set { P_0 , P_3 , P_6 , P_9 , P_{12} , P_{15} , P_{17} } of patients who undergo dialysis in the morning on the first day of the simulation.

Note that the infection source is randomly chosen from V'_p and the rest of the agents are susceptible, which allows the use of linearity of expectation to compute R_0 of the simulation:

$$R_0 = \frac{1}{|V_P'|} \sum_{u \in V_P' \lor \in V_H \cup V_P \setminus \{u\}} P_{u,v}$$

$$\tag{5}$$

Recall that this is the R_0 for the scenario in which the infection source is a dialysis patient who arrives for dialysis in the morning session of the first day.

In Eq (5), R_0 is a function of (*i*) the number of contacts $x_d(u, v)$ between agents *u* and *v* on a day *d*, as specified by the contact network, (*ii*) the daily shedding level s_d of an agent, and (*iii*) the scaling parameter α . Since the contact network is given and we have a target value of *f*, via Eq (5) we see that R_0 is a function of parameters α and γ .

For the final stage of our model calibration, let *T* denote the total volume of shedding of infectious individual. This volume can be expressed in terms of the parameter γ as follows.

$$T = \alpha \cdot \left(\frac{1}{1 - \frac{f}{100}}\right) \cdot \left(\sum_{j=0}^{6} \frac{1}{\gamma^{j}}\right)$$
(6)

Eqs 1, 5 and 6 express the three constraints our calibrated shedding models needs to satisfy. We use these equations to compute parameter values for our models as follows. Suppose that we have a target R_0 and two target shedding models *exp/exp* (f_1 %) and *exp/exp* (f_2 %).

To solve for the parameters of the model exp/exp (f_1 %), we pick an arbitrary value for parameter γ , denoted γ_1 . This fixes the value β_1 for parameter β as per Eq 1 and completely defines the shedding level s_d for all days d. We then use Eq 5 to solve for a value α_1 for parameter α . Since R_0 grows monotonically with α , to find an appropriate value of α that yields a target



Fig 3. Contact network G. HCP nodes, MWF patient nodes, and TThS patient nodes are depicted in blue, chocolate, and burlywood colors, respectively. Contacts within the same type of nodes are represented as edges colored according to node type. Grey colored edges represent HCP-patient contacts. The thickness of the edge corresponds to the contact duration. Patients mostly have contact with other patients at the same dialysis session unless there is an overlap between sessions, hence we observe 6 patient clusters (3 sessions per day on MWF and TThS) of 6 or 7 patients each. For any session, a patient will be in contact with at most 2 neighboring patients, thanks to how neighboring dialysis chairs are positioned. Over several sessions, depending on which chair they occupy, a patient may come in contact with several other patients.

 R_0 , we simply use binary search over the space of α . The triple (α_1 , β_1 , γ_1) provides a complete calibration of the shedding model *exp/exp* (f_1 %), yielding the target R_0 .

We next solve for a triple (α_2 , β_2 , γ_2) to calibrate model *exp/exp* (f_2 %) as follows. Since we want to preserve total shedding volume across shedding models *exp/exp* (f_1 %) and *exp/exp* (f_2 %), we use the total shedding volume for the shedding model *exp/exp* (f_1 %) as the left hand

	<i>exp/exp</i> (20%)			<i>exp/exp</i> (60%)		
R ₀	α	β	γ	α	β	γ
2.0	1.29e-3	3.01	2	1.009e-3	1.475	4.65
2.5	1.712e-3	3.01	2	1.355e-3	1.479	4.85
3.0	2.183e-3	3.01	2	1.723e-3	1.479	4.9

Table 1. Calibrated values of the	parameters for th	e shedding model.
-----------------------------------	-------------------	-------------------

side of Eq 6. This equation, along with Eq 5, constitute a system of two non-linear equations defined on variables α and γ . We solve for values α_2 and γ_2 using an iterative grid search which is aided by the fact that R_0 is monotonic in α (see Eq 5) and *T* is monotonic in both α and γ (see Eq 6). Given f_2 , we can use Eq 1 to compute the value β_2 of parameter β from γ_2 .

Using the above method, for $R_0 = 3$, we obtain values $\beta = 3.01$ and $\gamma = 2.0$ for the *exp/exp* (20%) shedding model and $\beta = 1.479$ and $\gamma = 4.90$ for the *exp/exp* (60%) model. These two shedding models are illustrated in Fig 2. The value of α computed for these two models is different, as reflected by the different peak shedding levels in the two models. However, the total shedding volume is the same (by design). The calibrated values of the parameters for the two shedding model are reported in Table 1.

NPIs

We used our simulator to explore the impact of six different NPIs, both individually as well as in certain plausible combinations. These six NPIs are described in the next six subsections.

Voluntary self-isolation. Voluntary self-isolation refers to the practice wherein a symptomatic HCP elects to stay home for 14 days starting the day after symptoms begin with voluntary isolation compliance rate $r_{VI} \in \{0.5, 0.7\}$. While in isolation, the HCP is replaced by a temporary HCP, selected from a community with COVID-19 attack rate of 0.347% (the COVID-19 attack rate in Johnson County, IA as of early-August 2020, computed by dividing active cases by the population) [44]. If the incoming replacement is exposed to COVID-19, we make the worst-case assumption that the substitute is in the first day of their presymptomatic period.

Active syndromic surveillance and compulsory isolation. Active syndromic surveillance refers to the practice of testing HCPs before their shift, and sending all HCPs with symptoms home. As with voluntary self-isolation, the isolation period is 14 days, and the HCP is replaced with a temporary HCP selected from the community. Unlike voluntary self-isolation, however, isolation in this case is not optional, so the compliance rate is by definition always 100%.

Using masks and/or N95 respirators. Surgical masks and N95 respirators interfere with the ability of an infected agent to transmit the virus to a susceptible agent. The difference between the two is the ability to attenuate transmission: our model attenuation factors are 0.93 (for N95 respirators) and 0.4 (for surgical masks) in accordance with results found in the literature [45, 46]. In this paper, we consider three plausible mask interventions: (*i*) patients wear surgical masks and HCPs are unmasked; (*ii*) both patients and HCPs wear surgical masks; and (*iii*) patients wear surgical masks and HCPs wear N95 respirators. We assume that all agents are 100% compliant, all the time.

Improved social distancing among HCPs. Limiting physical interactions is an effective intervention for reducing the transmission of COVID-19 [47]. Social distancing in the dialysis unit refers to the practice of HCPs staying at least 6 feet away from each other when not directly engaged in patient care. In our simulations, we model this form of social distancing by

removing a given fraction $r_{SD} \in \{0.25, 0.5, 0.75, 1\}$ of contacts that occur between pairs of HCPs when outside the immediate proximity of a dialysis station.

Increased physical separation between dialysis stations. Dialysis chairs are placed roughly 6 feet apart from each other in the UIHC dialysis unit, just at the outer limit of the generally accepted respiratory droplet range. As a result, infected patients may be close enough to infect patients in adjacent chairs, and, more to the point, HCPs attending to patients in adjacent chairs may be quite close to each other. In our simulations, we model increased physical separation between dialysis stations by removing a given fraction $r_{PS} \in \{0.25, 0.5, 0.75, 1\}$ of contacts between patients at adjacent dialysis chairs that are within 6 feet distance, and between HCPs that had contacts at adjacent dialysis stations.

Patient isolation and preemptive isolation of exposed HCPs. Many dialysis centers, like the UIHC, have an isolation room, where infectious patients may undergo dialysis without endangering others. For this intervention, we assume the first symptomatic patient in the unit will henceforth receive dialysis treatment in the isolation room. Preemptive isolation of exposed HCPs refers to the practice of placing the top k HCPs ($k \in \{1, \dots, 5\}$) having had the most exposure to that now isolated symptomatic patient in compulsory isolation, replacing them with temporary HCP selected from the community. Unlike active syndromic surveillance and compulsory isolation, isolation is being imposed here based on the HCPs likely increased risk of infection as opposed to any actual evidence of infection. For small values of k (e.g., k = 1, 2), it is possible to implement this intervention simply by assigning primary and/or secondary HCPs to each dialysis session with the understanding that the primary HCP will provide most care during the session and will be assisted by the secondary HCP as needed.

NPIs in combination. We hypothesize that combinations of low-cost versions of these interventions could more effectively reduce the attack rate: we therefore explore the following plausible combinations of interventions:

- Baseline+: Surgical masks for all HCPs and patients, improved social distancing among HCPs ($r_{SD} = 0.25$), and increased physical separation between dialysis stations ($r_{PS} = 0.75$).
- Baseline++: All Baseline+ interventions plus patient isolation and preemptive isolation of exposed HCPs (*k* = 1) upon detection of the first symptomatic patient.
- Baseline+++: All Baseline++ interventions plus use of N95 respirators by all HCPs for 2 weeks following detection of the first symptomatic patient.

Note that the extra interventions included in Baseline++ and Baseline+++ are triggered by the detection of the first symptomatic patient. This approach is an attempt to keep interventions targeted and their costs low.

Simulation setup

We create 6 models by independently using three different values of $R_0 \in \{2.0, 2.5, 3.0\}$ and two viral shedding models *exp/exp* (20%) and *exp/exp* (60%). In addition, we consider the following two initial conditions.

Scenario 1: One infected patient introduced during the first day's morning block; or

Scenario 2: One infected HCP introduced during the first day's morning block.

As a baseline, we run simulations for 30 consecutive days for each of these 12 settings. We then run 30-day simulations evaluating each of the 6 NPIs individually. Finally, we evaluate the three intervention bundles Baseline+, Baseline++, and Baseline+++ described above.

Our Python code to generate contact networks and run our simulations along with extensive documentation is available on our github page [48], for free download and use.

Results

Contact networks

We gathered ten days of HCP movement and interaction data. Six days had 14.5–15 hours of observation (days 2, 6, 7, 8, 9, and 10), whereas the remaining four days had an average of 6.5 hours of observation. The results presented in the paper are based on Day 10 interaction data. Results based on other days of data are provided in Figs W-AP in the <u>S1 Appendix</u>. Fig 3 and Table 2 depict the static contact network *G* and provide summary statistics, respectively. Table 3 shows statistics of the temporal contact networks, averaged over all *d* and *t*. The network statistics of *G* and $G^{d,t}$ reveal the high density of close range contacts among agents in the dialysis unit. For example, the average weighted degree of an HCP is 8.10 hrs per day (row 7 in Table 2), implying that on average an HCP in the dialysis unit has a total of 8.10 hrs of close contact with other dialysis unit agents in a day. Note that this quantity is high because agents typically have multiple simultaneous contacts. This is supported by the finding in Table 3 on temporal graphs $G^{d,t}$ that shows that on average each HCP is in contact with more than 1.6 other agents per timestep.

Baseline simulations

We turn our attention now to the results of our simulations, focusing on worst case modeling assumptions (i.e., $R_0 = 3.0$, and the *exp/exp* (60%) shedding model). We perform 500 replicates of each trial and report average values to account for stochastic effects in the simulation. Later, we summarize results obtained for both scenarios with different target R_0 values and shedding models.

Number of nodes	51				
Number of edges	520				
Other statistics	mean	std. dev.	max		
Degree	19.96	12.25	50		
Weighted degree (hrs per day)	3.77	2.66	11.12		
HCP degree	41.73	9.18	50		
HCP weighted degree (hrs per day)	8.10	2.72	11.12		
patient degree	13.98	1.44	17.00		
patient weighted degree (hrs per day)	2.59	0.65	4.19		
HCP-HCP edge weight (hrs per day)	0.41	0.26	1.11		
HCP-patient edge weight (hrs per day)	0.13	0.08	0.43		
patient-patient edge weight (hrs per day)	0.28	0.17	0.78		

Table 2. Network statistics of static graph G.

https://doi.org/10.1371/journal.pcbi.1009177.t002

Table 3. Network statistics of temporal graph $G^{d,t}$. The statistics shown here are obtained by averaging over all (approximately) 177K temporal graphs (177K \approx 26 days × 6822 timesteps per day).

	mean	std dev	max
overall degree	1.760	0.640	2.744
HCP degree	1.635	0.396	2.117
patient degree	1.744	0.619	2.536

https://doi.org/10.1371/journal.pcbi.1009177.t003



Fig 4. Cumulative distributions of transmission events over 30 days in Baseline simulation. (A) Scenario 1: dialysis patient is the infection source and (B) Scenario 2: HCP is the infection source. In both scenarios, the most frequent transmission route is from HCP \rightarrow patient (52.7% in Scenario 1 and 56.8% in Scenario 2), followed by patient \rightarrow patient (22.1% in Scenario 1 and 22.5% in Scenario 2), HCP \rightarrow HCP (16.6% in Scenario 1 and 18.5% in Scenario 2), and a small fraction of patient \rightarrow HCP transmission (6.7% in Scenario 1 and 2.6% in Scenario 2).

No interventions are imposed in the Baseline simulation: infected HCPs report for work as usual, and patients continue to dialyze in accordance with their original schedule even if infected. The high contact environment of the dialysis unit combined with the relatively high viral shedding rate of the shedding model ensures that almost everyone in the unit is infected within 30 days. The average (median) attack rates were 87.8% (94.1%) and 94.8% (96.1%) in Scenario 1 and Scenario 2 respectively. Furthermore, a substantial fraction of the simulations -45.8% and 53.6% in the two scenarios-resulted in a near 100% attack rate (attack rate of 96.1% or higher). The simulation also provides some insight into the underlying disease dynamics; this is shown in Fig 4. Independent of scenario and despite the fact that patients outnumber HCPs nearly 4:1, HCPs are the source of 69.3% of infections in Scenario 1 (infection source: dialysis patient; see Fig 4A) and 75.3% of the infections in Scenario 2 (infection source: HCP; see Fig 4B). We also note a surprisingly high number of patient \rightarrow patient transmissions, a consequence of the physical proximity of dialysis stations. In both scenarios, there are very few patient \rightarrow HCP transmissions. However, there are enough patient \rightarrow HCP transmissions in Scenario 1, early in the simulation, e.g., on average 2 HCPs get infected by patients in the first week, that then allows HCPs to become main source of transmissions. This implies that in Scenario 1, patient \rightarrow HCP transmissions might be the ones to most aggressively target via interventions. However, the situation in Scenario 2 is considerably more dire. In this scenario, almost immediately there are some HCP \rightarrow HCP transmissions and HCP \rightarrow patient transmissions. This means that by the time symptoms emerge in any infected agent, triggering the interventions in Baseline++ and Baseline+++, disease has already spread to a number of agents and is hard to subsequently contain.

Effect of individual NPIs

Figs <u>5A</u> and <u>6A</u> compare the cumulative attack rates for voluntary self-isolation ($r_{VI} \in \{0.5, 0.7\}$) as well as active syndromic surveillance and compulsory isolation against that of the Baseline strategy (no interventions). These interventions have only minimal effect on the overall attack rate, but at a cost of employing a large number of additional substitute HCPs (replacement HCP counts are shown in parenthesis). For example, using the active syndromic



Fig 5. Attack rates for individual NPIs and selected NPIs in combination in Scenario 1 (infection source: Dialysis patient), $R_0 = 3$, exp/exp (60%). (A) Voluntary self-isolation vs. active syndromic surveillance and compulsory isolation. (B) Use of masks and respirators. (C) Improved social distancing among HCPs. (D) Increased physical separation of dialysis stations. (E) Patient isolation and preemptive isolation of exposed HCPs. (F) Combinations of inexpensive NPIs denoted Baseline+, Baseline++, and Baseline+++. Each figure displays cumulative attack rates as a function of simulation days. Each plot includes the Baseline curve for comparison; lower curves correspond to fewer infected agents. See text for descriptions of the individual NPIs.



Fig 6. Attack rates for individual NPIs and selected NPIs in combination in Scenario 2 (infection source: HCP), $R_0 = 3$, *exp/exp* (60%). These are the same plots as in Fig 5, except that these are for Scenario 2.

surveillance and compulsory isolation intervention requires 10.0 additional substitute HCPs in Scenario 1 (see Fig 5A) and 11.3 in Scenario 2 (see Fig 6A). In Table A in S1 Appendix, we provide additional analyses on the voluntary self-isolation.

Figs 5B and 6B compares the cumulative attack rates when using masks and/or respirators with the Baseline strategy (no interventions) in Scenario 1 and 2 respectively. The three curves shown correspond to (*i*) patients wear surgical masks and HCPs are unmasked (78.0% and 93.0% overall attack rate in the two scenarios); (*ii*) patients wear surgical masks and HCPs wear surgical masks (70.0% and 84.7% overall attack rate in the two scenarios); and (*iii*) patients wear surgical masks and HCPs wear surgical masks and HCPs wear N95 respirators (22.4% and 16.8% overall attack rate in the two scenarios). Patients wearing surgical masks and HCPs wearing N95 respirators throughout their time in the dialysis unit has a significant effect on overall attack rate, leading to a 74.4% and 82.3% reduction in the two scenarios.

Figs 5C and 6C compare the cumulative attack rates observed with increasing social distancing among HCPs, again, in comparison with the Baseline strategy (no interventions). Recall the underlying model for this intervention is to remove a fraction $r_{SD} \in \{0.25, 0.5, 0.75, 1\}$ of contacts that occur between pairs of HCPs when outside the immediate proximity of a dialysis station, while leaving HCP/patient and patient/patient contacts unchanged. These interventions do not have a significant effect on overall attack rates unless these pairs of HCP/ HCP contacts get removed completely.

Increasing the physical separation between dialysis stations (Figs 5D and 6D) reduces opportunities for transmission not only between patients in adjacent chairs, but also with HCPs directly attending to either of the two stations. Recall that increasing the physical separation between dialysis stations is modeled by a parameter $r_{PS} \in \{0.25, 0.5, 0.75, 1\}$ that denotes the fraction of relevant contacts that are removed. When compared with the Baseline strategy (no interventions), we see some reduction in overall attack rate in Scenario 1 (infection source: dialysis patient) as the degree of additional separation increases; complete separation reduces the Baseline strategy's 87.8% overall attack rate to 74.5%, although physical space constraints may preclude achieving complete separation in any real application. We see a similar effect in Scenario 2, but the magnitude of the reduction is a little bit smaller compared to that of Scenario 1.

Figs 5E and 6E compare the effect of isolating the first patient with detectable symptoms and preemptively isolating the top k HCPs ($k \in \{1 \dots 5\}$) exposed to that patient with the Baseline strategy (no interventions). In Scenario 1 (infection source: dialysis patient), as the figure makes clear, the majority of the effect of this intervention occurs with k = 1, that is, preemptive isolation of the single most exposed HCP, which reduces the attack rate to 75.9%. In contrast, in Scenario 2 this intervention has no noticeable effect.

Combining NPIs

Figs 5F and 6F compare three strategies that combine multiple NPIs with the Baseline strategy (no interventions). Because the combined interventions shown here (Baseline+, Baseline++, and Baseline+++) consist of increasingly "nested" sets of interventions, it is not entirely surprising that each strategy demonstrates increasing advantage, measured in terms of cumulative attack rates, with respect to the Baseline.

The Baseline+ strategy (surgical masks for HCPs and patients; improved social distancing among HCPs r_{SD} = 0.25; and increased physical separation between dialysis stations r_{PS} = 0.75) reduces the attack rate to 56.2% in Scenario 1 and to 75.8% in Scenario 2 from a Baseline attack rate of 87.8% and 94.8% in the two scenarios. The Baseline++ strategy, which adds patient isolation and preemptive isolation of a single exposed HCP to the Baseline+ strategy, reduces the



Fig 7. Frequency of replicates as a function of attack rates for different NPIs: (A) Scenario 1 and (B) Scenario 2. Each histogram displays the number of replicates (y axis; out of 500) with the specified overall attack rate. In Scenario 1, two peaks are commonly observed for each histogram; one where no transmission events occur (the infection is immediately quashed) and one indicating the fuller extent of an eventual outbreak. When no intervention is imposed (Baseline), only 31 simulations (6.2%) have no transmission events, while 83.6% (418) resulted in attack rates larger than 90%. In contrast, Baseline+++ and Baseline+++ significantly increases the likelihood of observing no transmission events (162 and 175 simulations, 32.4% and 35.0%, respectively). In Scenario 2, even with interventions, infection is never quashed in any replicate. However, even in this scenario, there are no replicates with Baseline+++ interventions that have a near-100% attack rate. Furthermore, the frequency mass of the Baseline+++ intervention is substantially shifted to the left (towards lower attack rates) relative to the Baseline simulations.

attack rate to 46.5% in Scenario 1, but yields no improvement in the attack rate relative to Baseline+ in Scenario 2. Finally, Baseline+++, which adds 2 weeks of N95 usage for HCPs following the detection of the first symptomatic patient, produces an impressive 60.4% reduction in the Baseline strategy's attack rate in Scenario 1 (to 34.8%), at a relatively modest cost of 0.7 replacement HCPs and a similarly modest number of N95 masks for HCPs only over a two week period. Moreover, unlike with the Baseline++ strategy, the Baseline+++ strategy also has a strong positive effect in Scenario 2, reducing the attack rate to 56.7% at a small cost of 1 replacement HCP.

Reporting the average attack rates obscures the distinction between replicates where the infection fails to spread beyond the first infected agent and those where the infection permeates the agent population. The histograms in Fig 7 makes the nature of this mixture explicit by showing the number of replicates (y axis; out of 500 total) that result in a given attack rate (x axis). What the plot for Scenario 1 (Fig 7A) makes clear is that more effective strategies (here, Baseline \ll Baseline+ \ll Baseline++ \ll Baseline+++) affect both aspects of performance: it is both more likely that the infection is immediately quashed and that—when it does spread—the spread is decidedly more limited. So for the Baseline strategy, 83.6% of replicates (418) had attack rates larger than 90%, while only 6.2% of replicates (31) had no transmission events. Successively more effective strategies increased the number of quashed infections to a greater degree, with 32.4% of the Baseline+++ replicates (162) having no transmission events at all. Furthermore, the frequency mass of each strategy with interventions moved substantially to the left, relative to the immediately less effective strategy. In Scenario 2 (Fig 7B), the situation is not as positive mainly because despite the interventions, the infection is never quashed immediately. Some of the interventions that are effective in Scenario 1 (e.g., patient isolation and preemptive isolation) start too late for Scenario 2 and others (e.g., increased physical separation of dialysis units) have too little effect, given how many agents quickly become infected. Nevertheless, the frequency mass of the Baseline+ and Baseline++ strategies moved substantially to the left relative to the

Baseline strategy and the frequency mass of the Baseline+++ strategy moved even more to the left relative to these less comprehensive strategies.

In Figs S-V in the S1 Appendix, we show all the above-mentioned results for the *exp/exp* (20%) shedding model, for both scenarios, and for $R_0 = 3.0$. The symptom-based interventions (e.g. voluntary self-isolation, active syndromic surveillance, and isolating the first patient with detectable symptoms and preemptively isolating top *k* exposed HCPs) are more effective in *exp/exp* (20%) shedding model, because the total amount of shedding before symptom is less compared to the *exp/exp* (60%) shedding model. In Figs A-R in the S1 Appendix, we report results for the *exp/exp* (60%) shedding model for other values of R_0 , i.e., $R_0 \in \{2.0, 2.5\}$. We also report results for the *exp/exp* (60%) shedding model with $R_0 = 3.0$, using other values, 1.5 and 2.5, of the shedding ramp down parameter γ .

Table 4 summarizes attack rates for Baseline, Baseline+, Baseline++, and Baseline+++ for both scenarios, both shedding models, and three different target R_0 values. As the combined interventions become increasingly sophisticated, the attack rates for Scenario 1 consistently drop regardless of shedding model and the target R_0 value. The attack rates in Scenario 2, where the infection source is the HCP, were larger than their corresponding attack rates in Scenario 1. The attack rates of same set of interventions on the *exp/exp* (60%) shedding model is always larger than that of the *exp/exp* (20%) shedding model.

As a form of validation, we ran the same set of simulations using HCP movement and interaction patterns extracted from the other days data within our dataset. These are also reported in Figs W-AP in the <u>S1 Appendix</u>. And while the attack rates varied slightly due to minor differences in the number of HCPs and patients as well as their underlying contact patterns, we observed the same relationship between intervention strategy and reduced attack rates as we moved from the Baseline strategy towards Baseline+++.

Discussion

Evidence suggests that viral respiratory infections like COVID-19 can spread very quickly in densely populated environments. For example, in early March 2020, 94 workers in a South Korean call center (43.5% of the total) were infected, notwithstanding that immediate action was taken to close the call center when the outbreak was first reported [49]. Dialysis units are also good examples of densely populated environments, where frequent close contacts occur primarily between patients and HCPs on a recurring basis. Morover, not only are dialysis settings particularly suited to the transmission of COVID-19, but dialysis patients are particularly susceptible to adverse effects of the infection thanks to their underlying health concerns. A recent study examining clinical course and outcomes for 36 COVID-19 positive hemodialysis patients in Spain found that 18 of the patients eventually suffered degraded clinical status and 11 patients died outright, a 30.5% mortality rate [12]. Our simulations support this evidence

SM	R ₀	Scenario 1			Scenario 2				
		В	B+	B++	B+++	В	B+	B++	B+++
	2.0	62%	29%	19%	15%	82%	52%	51%	36%
20%	2.5	75%	45%	31%	22%	88%	65%	64%	46%
	3.0	84%	50%	41%	29%	93%	74%	72%	54%
	2.0	69%	32%	24%	17%	84%	54%	54%	39%
60%	2.5	83%	48%	37%	28%	91%	66%	65%	49%
	3.0	88%	56%	47%	35%	95%	76%	74%	57%

Table 4. Attack rates for combinations of NPIs under different modeling assumptions. "SM" is short for "shedding models", "B" is short for "Baseline," in Scenario 1 the infection source is a dialysis patient and in Scenario 2 the infection source is a HCP.

https://doi.org/10.1371/journal.pcbi.1009177.t004

and show how a single infection in a dialysis unit can quickly lead to nearly universal infection across the unit due largely to the intensity of contacts among HCPs and patients.

In summary, in both scenarios, the most effective individual NPI is the continuous use of surgical masks by patients and N95 respirators by HCPs. In Scenario 1, two other interventions, namely (i) increased physical separation between dialysis stations and (ii) patient isolation and preemptive isolation of a few exposed HCPs are moderately successful, whereas in Scenario 2, (i) increased physical separation between dialysis stations and (ii) social distancing among HCPs are somewhat effective.

The relatively small impact on attack rates for voluntary self-isolation and active syndromic surveillance and compulsory isolation is likely attributable to the high rate of viral shedding prior to emergence of symptoms in the *exp/exp* (60%) shedding model. As we experience shortages of available HCPs in the midst of the COVID-19 outbreak [50], substituting a large number of HCPs may not be a viable option, especially given the relatively low reduction in attack rate. As a result, we conclude that voluntary self-isolation and active syndromic surveillance and compulsory isolation are not particularly effective NPIs in the dialysis unit.

The usage of masks, especially for the case where patients wear surgical masks and HCPs wear N95 respirators, is substantially better than the rest of the NPIs. If the continuous use of N95 respirators is feasible, then the consistent use of surgical masks for patients and N95 respirators for HCPs is clearly an intervention that should be implemented. Unfortunately, due to the sharp increase in demand for N95 respirators [51], it in unclear if this is a feasible intervention at a typical dialysis unit. This observation inspired our Baseline+++ strategy, which includes a more limited deployment of N95 respirators once the first symptomatic patient is detected (see Section "Combining NPIs").

The strategy of isolating the first patient with detectable symptoms followed by isolating preemtive isolation of exposed HCPs performs well in Scenario 1. Replacing more HCPs had relatively little effect on the attack rate, while concomitantly raising the cost in terms of additional HCPs, which, as noted previously, may not be feasible. Moreover, with this intervention it is at least technically possible that one might isolate a healthy HCP while importing an infected substitute. Notice that this intervention had no noticeable effect in Scenario 2. This can be explained by the transmission dynamics shown in Fig 4. By the time the first patient shows any symptoms, the disease has spread from the infection source (an HCP) to several other agents. At this point, isolating the first patient with detectable symptoms and preemptively isolating a few HCPs is "too little, too late."

The primary message of this paper is that a strategy combining simple, inexpensive, and practical interventions can substantially reduce the overall attack rate. The Baseline+ strategy (surgical masks for HCPs and patients; improved social distancing among HCPs $r_{SD} = 0.25$; and increased physical separation between dialysis stations $r_{PS} = 0.75$) shows substantial reduction in the attack rate in both scenarios. However, the fact that the attack rate is not reduced as much in Scenario 2 could be due to the higher baseline attack rate observed in Scenario 2. In the Baseline++ strategy (Baseline+; patient isolation and preemptive isolation of a single exposed HCP), the difference between the impressive reduction in attack rate observed in Scenario 1 and the negligible effect observed in Scenario 2 is likely due to the fact that patient isolation and preemptive isolation of a single exposed HCP is "too little, too late" given how much the disease has spread before the first patient shows symptoms.

The attack rates in Scenario 2 were larger than their corresponding attack rates in Scenario 1, perhaps indicating the initial patient-to-HCP jump in Scenario 1 presents a bottleneck to broader transmission. Supporting this particular intuition is the fact that, in Scenario 2, attack rates for Baseline++ are similar to those of Baseline+, indicating that isolating the first symptomatic patient and preemptive isolation the corresponding most exposed HCP does not have

much effect. The attack rates in exp/exp (60%) shedding model were larger than their corresponding attack rates of the same set of interventions in exp/exp (20%) shedding model. Although the volume of shedding is set to be the same for the two shedding models, the difference in the dynamics comes from the percentage of presymptomatic shedding.

The Baseline+++ intervention (Baseline++; use of N95 respirators by all HCPs for 2 weeks following detection of the first symptomatic patient) performed well in the simulations reported here. Note that the primary interventions in Baseline+++ are all relatively inexpensive by design: relatively more expensive interventions (such as preemptive HCP isolation, or the use of N95 respirators) are applied in a targeted fashion only when an infection is first detected. Such targeting helps to reduce overall cost and make the Baseline+++ strategy economically feasible. Implementing Baseline+++ in dialysis unit modeled here would, in practice, require one additional temporary HCP for 12 days (Sundays excluded) and 132 N95 respirators are needed (11 HCPs, 12 days).

A second interesting finding, clearly evident in Fig 7, is that an infection in the dialysis unit can evolve in two distinct ways: it is either immediately suppressed, or it eventually overtakes nearly the entire unit. Once the infection spreads to other agents, even the combination of interventions in the Baseline+++ strategy cannot effectively dampen the outbreak. This observation highlights the critical temporal aspects involved in mounting an adequate response, and suggests that rigorous but targeted practices applied to prevent subsequent transmission beyond the first patient may be the key to preventing a catastrophic outbreak of COVID19 in closed, high-contact systems like our dialysis unit. Here, we focused on cost-efficient interventions that can be easily implemented in practice. However, if COVID-19 testing were a viable option in terms of cost, an even more rigorous program of targeted interventions could be deployed with confirmed COVID-19 patients. Indeed, a recent study of a COVID-19 outbreak in a dialysis center (Wuhan, China; January-March 2020) provides some evidence to support this claim [52]. The study reports that 18.26% of dialysis patients and 12.12% of HCPs in their facility were diagnosed with COVID-19. This attack rate is much lower than in our simulations, but we note that they followed a rigorous program of interventions guided by an aggressive in-house COVID-19 testing program. Within one week of the first confirmed COVID-19 case, all HCPs wore full protective gear, including waterproof disposable gowns, caps, gloves, face shields, and N95 respirators. In addition, an extensive cleaning and disinfection program was implemented. Finally, during the screening period, all infected patients and HCPs were isolated or transferred to other hospitals.

In our modeling and simulations, we used R_0 values in the range [2, 3]. These were based on estimates of R_0 for COVID-19 in the general population. There are almost no prior studies that can be used to estimate an R_0 value or an attack rate in a dialysis unit. One exception is a study that performed contact tracing of a COVID-19 outbreak at a single hemodialysis center in Toronto, Canada [53]. There were two seed cases, a patient and a HCP. The seed patient infected one other patient and the seed HCP infected six others (4 patients and 2 HCPs). This can be interpreted as yielding an average R_0 of 3.5 (2 seeds leading to 7 secondary cases). This trend—of an infected patient leading to fewer secondary cases compared to an infected HCP —is what we observe in our simulations as well. The overall attack rate reported in this study was low (0.07), due to an extensive intervention upon detection of a few COVID-19 cases. These interventions included universal droplet and contact precautions including gloves, face shields, surgical masks, and isolation gowns, that were imposed on top of the interventions that were already in place (e.g., physical distancing, universal masking for staff, and screening for symptoms for patients).

In addition to COVID-19, dialysis patients are at risk for other respiratory infections spread by droplets. Accordingly, our results have implications for how to prevent the spread of other

respiratory infections. For example, the acquisition of MERS, a virus similar to COVID-19, was also associated with treatment in dialysis centers [54, 55]. In addition, every year, hemodialysis patients are at risk for acquiring influenza, and they are more likely to suffer from worse influenza-associated outcomes than patients not on hemodialysis. Also, they are less likely to benefit from the influenza vaccine than people not undergoing hemodialysis [16].

Some of our results also have implications for healthcare-associated infections spread by close contact (e.g., multi-drug-resistant infections spread by the hands of healthcare workers). Specifically, patients on hemodialysis are at risk for a wide range of multidrug resistant organisms (MDROs) including MRSA, vancomycin-resistant Enterococcus (VRE), and carbape-nem-resistant Enterobacteriaceae (CRE) [3] and MDRO colonization rates are extremely high [56–59].

A comprehensive study on the comparison of dialysis facilities showed that the mean of the number of stations over 5,193 facilities was 18.9 (std dev 7.8) [60]. So the UIHC dialysis center is somewhat smaller than average. Although there are larger and smaller dialysis units, the demands of care require a specific and fairly constant distribution of interactions with members of the healthcare staff on average over a typical shift. For instance, there was a small variance in the dialysis sessions of 3.8 (0.3) hours across all facilities as well as the number of dialysis sessions per week per patient 3.0 (0.2). (see Table 4.1 from [60]). As far as we know, dialysis centers have a similar physical layout: nurses' station in the central area and the stations placed around the outer perimeter of the unit. Even if dialysis units have radically different layout, HCP-patient contact patterns and patient-patient contact patterns will likely remain unchanged. HCP-HCP contact patterns may change, but as seen in Fig 4 HCP-patient contacts and patient-patient contacts play a dominant role in transmissions. The fact that COVID-19 is primarily spread by airborne respiratory droplets also means that the physical arrangement of surfaces in the dialysis unit does not play as much of a role as it might in the spread of an infection such as MRSA whose transmission is fomite-based [20, 61, 62]. Thus, we anticipate that our results and findings based on our simulations should be generalizable to other dialysis settings.

Our results also have broad applications for other healthcare settings beyond dialysis centers where patients repeatedly receive care for prolonged periods. Specific examples include chemotherapy units where patients come regularly and sit in chairs for several hours at a time. In addition, patients receiving parenteral antibiotics through peripherally inserted central catheter (PICC) lines arrive daily and can stay up to several hours, receiving monitoring and care from healthcare professionals. Like dialysis, for both chemotherapy and parenteral antibiotic treatment, the longest durations of interactions between healthcare professionals and the patients under their care occur at the beginning and end of a therapy session.

Our work is subject to a few limitations. First, although we have explored interventions that reduce HCP-HCP contacts and patient-patient contacts, we did not explore interventions that reduce or modify HCP-patient contacts. Since most of these result from direct patient care activity, it would not be reasonable to expect a uniform reduction in these contacts as we did for the other types of contacts. We could, however, consider reorganizing dialysis schedules so as to reduce the diversity of HCP-patient contacts by, *e.g.*, assigning each patient to a unique HCP in order to reduce redundant transmission paths. Second, we have limited our model to consider only person-to-person respiratory droplet transmission of the virus. However, aero-sol and fomite transmission are also plausible, as SARS-CoV-2 can survive in aerosols for several hours and on surfaces for several days [63]. Detailed modeling of these transmission routes will require more data, such as viral survival rates in the air or on different types of surfaces. But these other transmission routes will also almost certainly suggest other interventions such as improvements in hand-hygiene protocols or additional cleaning practices which may

in turn become important elements of new cost-effective practical interventions. Third, we have limited our focus to transmissions within the dialysis unit. A recent study pointed out that sharing health-care transportation to the dialysis unit is a significant risk factor for COVID-19 infection [64]. Our paper does not consider the role of such interactions.

Supporting information

S1 Appendix. A **document that contains results on additional experiments.** In the paper, we showed simulation results focusing on the worst case modeling assumptions (i.e., $R_0 = 3.0$, and the *exp/exp* (60%) shedding model). Here, in the Appendix, we show results on other settings of the parameters as follows:

- *exp/exp* (60%) shedding model, $R_0 \in \{2.0, 2.5\}$
- *exp/exp* (60%) shedding model, $R_0 = 3.0, \gamma \in \{1.5, 2.5\}$
- exp/exp (20%) shedding model, $R_0 = 3.0$
- · Additional analyses on voluntary self-isolation
- exp/exp (60%) shedding model, $R_0 = 3.0$, on the other days of data

Fig A. Cumulative distributions of transmission events over 30 days in Baseline simulation on $R_0 = 2.0$, and the *exp/exp* (60%). (A) Scenario 1: dialysis patient is the infection source. (B) Scenario 2: HCP is the infection source. Fig B. Frequency of replicates as a function of attack rates for different NPIs on $R_0 = 2.0$, and the *exp/exp* (60%). (A) Scenario 1. (B) Scenario 2. Fig C. Attack rates for individual NPIs and selected NPIs in combination in Scenario 1 (infection source: dialysis patient), $R_0 = 2.0$, exp/exp (60%). (A) Voluntary self-isolation vs. active syndromic surveillance and compulsory isolation. (B) Use of masks and respirators. (C) Improved social distancing among HCPs. (D) Increased physical separation of dialysis stations. (E) Patient isolation and preemptive isolation of exposed HCPs. (F) Combinations of inexpensive NPIs. Fig D. Attack rates for individual NPIs and selected NPIs in combination in Scenario 1 (infection source: HCP), $R_0 = 2.0$, exp/exp (60%). (A) Voluntary self-isolation vs. active syndromic surveillance and compulsory isolation. (B) Use of masks and respirators. (C) Improved social distancing among HCPs. (D) Increased physical separation of dialysis stations. (E) Patient isolation and preemptive isolation of exposed HCPs. (F) Combinations of inexpensive NPIs. Fig E. Cumulative distributions of transmission events over 30 days in **Baseline simulation on** $R_0 = 2.5$, and the *exp/exp* (60%). (A) Scenario 1: dialysis patient is the infection source. (B) Scenario 2: HCP is the infection source. Fig F. Frequency of replicates as a function of attack rates for different NPIs on $R_0 = 2.5$, and the exp/exp (60%). (A) Scenario 1. (B) Scenario 2. Fig G. Attack rates for individual NPIs and selected NPIs in combination in Scenario 1 (infection source: dialysis patient), $R_0 = 2.5$, exp/exp (60%). (A) Voluntary self-isolation vs. active syndromic surveillance and compulsory isolation. (B) Use of masks and respirators. (C) Improved social distancing among HCPs. (D) Increased physical separation of dialysis stations. (E) Patient isolation and preemptive isolation of exposed HCPs. (F) Combinations of inexpensive NPIs. Fig H. Attack rates for individual NPIs and selected NPIs in combination in Scenario 1 (infection source: HCP), $R_0 = 2.5$, exp/exp (60%). (A) Voluntary self-isolation vs. active syndromic surveillance and compulsory isolation. (B) Use of masks and respirators. (C) Improved social distancing among HCPs. (D) Increased physical separation of dialysis stations. (E) Patient isolation and preemptive isolation of exposed HCPs. (F) Combinations of inexpensive NPIs. Fig I. Viral shedding models. The bar charts show the shedding level of an infected individual whose presymptomatic period and symptomatic

period are 6 and 7 days, respectively. The percentage in the parenthesis of each model corresponds to the percentage of shedding during the presymptomatic state (P) relative to the total shedding volume. The peak shedding level differs in the two models, but the total shedding volume remains the same. The shedding models are calibrated by setting the shedding ramp down parameter of the *exp/exp* (20%) shedding model $\gamma = 1.5$. Fig J. Cumulative distributions of transmission events over 30 days in Baseline simulation on $R_0 = 3.0$, and the exp/exp (60%) shedding model where $\gamma = 1.5$. (A) Scenario 1: dialysis patient is the infection source. (B) Scenario 2: HCP is the infection source. Fig K. Frequency of replicates as a function of attack rates for different NPIs on $R_0 = 3.0$, and the *exp/exp* (60%) shedding model where $\gamma = 1.5$. (A) Scenario 1. (B) Scenario 2. Fig L. Attack rates for individual NPIs and selected NPIs in combination in Scenario 1 (infection source: dialysis patient), $R_0 = 3.0$, exp/exp(60%), where $\gamma = 1.5$. (A) Voluntary self-isolation vs. active syndromic surveillance and compulsory isolation. (B) Use of masks and respirators. (C) Improved social distancing among HCPs. (D) Increased physical separation of dialysis stations. (E) Patient isolation and preemptive isolation of exposed HCPs. (F) Combinations of inexpensive NPIs. Fig M. Attack rates for individual NPIs and selected NPIs in combination in Scenario 1 (infection source: HCP), $R_0 = 3.0$, exp/exp (60%), where $\gamma = 1.5$. (A) Voluntary self-isolation vs. active syndromic surveillance and compulsory isolation. (B) Use of masks and respirators. (C) Improved social distancing among HCPs. (D) Increased physical separation of dialysis stations. (E) Patient isolation and preemptive isolation of exposed HCPs. (F) Combinations of inexpensive NPIs. Fig N. Viral shedding models. The bar charts show the shedding level of an infected individual whose presymptomatic period and symptomatic period are 6 and 7 days, respectively. The percentage in the parenthesis of each model corresponds to the percentage of shedding during the presymptomatic state (*P*) relative to the total shedding volume. The peak shedding level differs in the two models, but the total shedding volume remains the same. The shedding models are calibrated by setting the shedding ramp down parameter of the *exp/exp* (20%) shedding model γ = 2.5. Fig O. Cumulative distributions of transmission events over 30 days in Baseline simulation on $R_0 = 3.0$, and the exp/exp (60%) shedding model where $\gamma = 2.5$. (A) Scenario 1: dialysis patient is the infection source. (B) Scenario 2: HCP is the infection source. Fig P. Frequency of replicates as a function of attack rates for different NPIs on $R_0 = 3.0$, and the *exp/exp* (60%) shedding model where $\gamma = 2.5$. (A) Scenario 1. (B) Scenario 2. Fig Q. Attack rates for individual NPIs and selected NPIs in combination in Scenario 1 (infection source: dialysis patient), $R_0 = 3.0$, exp/exp (60%), where $\gamma = 2.5$. (A) Voluntary self-isolation vs. active syndromic surveillance and compulsory isolation. (B) Use of masks and respirators. (C) Improved social distancing among HCPs. (D) Increased physical separation of dialysis stations. (E) Patient isolation and preemptive isolation of exposed HCPs. (F) Combinations of inexpensive NPIs. Fig R. Attack rates for individual NPIs and selected NPIs in combination in Scenario 1 (infection source: HCP), $R_0 = 3.0$, exp/exp (60%), where $\gamma = 2.5$. (A) Voluntary self-isolation vs. active syndromic surveillance and compulsory isolation. (B) Use of masks and respirators. (C) Improved social distancing among HCPs. (D) Increased physical separation of dialysis stations. (E) Patient isolation and preemptive isolation of exposed HCPs. (F) Combinations of inexpensive NPIs. Fig S. Cumulative distributions of transmission events over 30 days in Baseline simulation on $R_0 = 3.0$, and the *exp/exp* (20%) shedding model. (A) Scenario 1: dialysis patient is the infection source. (B) Scenario 2: HCP is the infection source. Fig T. Frequency of replicates as a function of attack rates for different **NPIs on** R_0 = 3.0, and the *exp/exp* (20%) shedding model. (A) Scenario 1. (B) Scenario 2. Fig U. Attack rates for individual NPIs and selected NPIs in combination in Scenario 1 (infection source: dialysis patient), $R_0 = 3.0$, exp/exp (20%). (A) Voluntary self-isolation vs. active syndromic surveillance and compulsory isolation. (B) Use of masks and respirators. (C)

Improved social distancing among HCPs. (D) Increased physical separation of dialysis stations. (E) Patient isolation and preemptive isolation of exposed HCPs. (F) Combinations of inexpensive NPIs. Fig V. Attack rates for individual NPIs and selected NPIs in combination in Scenario 1 (infection source: HCP), R₀ = 3.0, exp/exp (20%). (A) Voluntary self-isolation vs. active syndromic surveillance and compulsory isolation. (B) Use of masks and respirators. (C) Improved social distancing among HCPs. (D) Increased physical separation of dialysis stations. (E) Patient isolation and preemptive isolation of exposed HCPs. (F) Combinations of inexpensive NPIs. Fig W. Contact network G on Day 2. We gathered ten days of HCP movement and interaction data where six days had 14.5-15 hours of observation (Day 2, Day 6, Day 7, Day 8, Day 9, and Day 10). HCP nodes, MWF patient nodes, and TThS patient nodes are depicted in blue, chocolate, and burlywood colors, respectively. We observe a total population of 42 agents (14 HCPs and 28 patients). Contacts within the same type of nodes are represented as edges colored according to node type. Grey colored edges represent HCP-patient contacts. The thickness of the edge corresponds to the contact duration. Fig X. Cumulative distributions of transmission events over 30 days in Baseline simulation on Day 2 and $R_0 =$ **3.0**. (A) the *exp/exp* (20%) shedding model on Scenario 1. (B) the *exp/exp* (20%) shedding model on Scenario 2. (C) the exp/exp (60%) shedding model on Scenario 1. (D) the exp/exp (60%) shedding model on Scenario 2. Fig Y. Attack rates for combinations of inexpensive **NPIs on Day 2**, $R_0 = 3.0$. (A) the *exp/exp* (20%) shedding model on Scenario 1. (B) the *exp/exp* (20%) shedding model on Scenario 2. (C) the *exp/exp* (60%) shedding model on Scenario 1. (D) the exp/exp (60%) shedding model on Scenario 2. Fig Z. Frequency of replicates as a function of attack rates for combinations of inexpensive NPIs on Day 2, $R_0 = 3.0$. (A) the exp/exp (20%) shedding model on Scenario 1. (B) the exp/exp (20%) shedding model on Scenario 2. (C) the exp/exp (60%) shedding model on Scenario 1. (D) the exp/exp (60%) shedding model on Scenario 2. Fig AA. Contact network G on Day 6. We gathered ten days of HCP movement and interaction data where six days had 14.5-15 hours of observation (Day 2, Day 6, Day 7, Day 8, Day 9, and Day 10). HCP nodes, MWF patient nodes, and TThS patient nodes are depicted in blue, chocolate, and burlywood colors, respectively. We observe a total population of 47 agents (13 HCPs and 34 patients). Fig AB. Cumulative distributions of transmission events over 30 days in Baseline simulation on Day 6 and $R_0 = 3.0$. (A) the *exp/exp* (20%) shedding model on Scenario 1. (B) the *exp/exp* (20%) shedding model on Scenario 2. (C) the *exp/exp* (60%) shedding model on Scenario 1. (D) the *exp/exp* (60%) shedding model on Scenario 2. Fig AC. Attack rates for combinations of inexpensive NPIs on Day 6, $R_0 =$ 3.0. (A) the exp/exp (20%) shedding model on Scenario 1. (B) the exp/exp (20%) shedding model on Scenario 2. (C) the exp/exp (60%) shedding model on Scenario 1. (D) the exp/exp (60%) shedding model on Scenario 2. Fig AD. Frequency of replicates as a function of attack rates for combinations of inexpensive NPIs on Day 6, $R_0 = 3.0$. (A) the *exp/exp* (20%) shedding model on Scenario 1. (B) the exp/exp (20%) shedding model on Scenario 2. (C) the exp/ exp (60%) shedding model on Scenario 1. (D) the exp/exp (60%) shedding model on Scenario 2. Fig AE. Contact network G on Day 7. We gathered ten days of HCP movement and interaction data where six days had 14.5-15 hours of observation (Day 2, Day 6, Day 7, Day 8, Day 9, and Day 10). HCP nodes, MWF patient nodes, and TThS patient nodes are depicted in blue, chocolate, and burlywood colors, respectively. We observe a total population of 46 agents (10 HCPs and 36 patients). Fig AF. Cumulative distributions of transmission events over 30 days in Baseline simulation on Day 7 and $R_0 = 3.0$. (A) the *exp/exp* (20%) shedding model on Scenario 1. (B) the exp/exp (20%) shedding model on Scenario 2. (C) the exp/exp (60%) shedding model on Scenario 1. (D) the exp/exp (60%) shedding model on Scenario 2. Fig AG. Attack rates for combinations of inexpensive NPIs on Day 7, $R_0 = 3.0$. (A) the *exp/exp* (20%) shedding model on Scenario 1. (B) the *exp/exp* (20%) shedding model on Scenario 2.

(C) the exp/exp (60%) shedding model on Scenario 1. (D) the exp/exp (60%) shedding model on Scenario 2. Fig AH. Frequency of replicates as a function of attack rates for combinations of inexpensive NPIs on Day 7, $R_0 = 3.0$. (A) the exp/exp (20%) shedding model on Scenario 1. (B) the exp/exp (20%) shedding model on Scenario 2. (C) the exp/exp (60%) shedding model on Scenario 1. (D) the exp/exp (60%) shedding model on Scenario 2. Fig AI. Contact network G on Day 8. We gathered ten days of HCP movement and interaction data where six days had 14.5-15 hours of observation (Day 2, Day 6, Day 7, Day 8, Day 9, and Day 10). HCP nodes, MWF patient nodes, and TThS patient nodes are depicted in blue, chocolate, and burlywood colors, respectively. We observe a total population of 45 agents (11 HCPs and 34 patients). Fig AJ. Cumulative distributions of transmission events over 30 days in Baseline simulation on Day 8 and $R_0 = 3.0$. (A) the *exp/exp* (20%) shedding model on Scenario 1. (B) the exp/exp (20%) shedding model on Scenario 2. (C) the exp/exp (60%) shedding model on Scenario 1. (D) the *exp/exp* (60%) shedding model on Scenario 2. Fig AK. Attack rates for combinations of inexpensive NPIs on Day 8, $R_0 = 3.0$. (A) the *exp/exp* (20%) shedding model on Scenario 1. (B) the exp/exp (20%) shedding model on Scenario 2. (C) the exp/exp (60%) shedding model on Scenario 1. (D) the exp/exp (60%) shedding model on Scenario 2. Fig AL. Frequency of replicates as a function of attack rates for combinations of inexpensive NPIs on Day 8, $R_0 = 3.0$. (A) the *exp/exp* (20%) shedding model on Scenario 1. (B) the *exp/exp* (20%) shedding model on Scenario 2. (C) the *exp/exp* (60%) shedding model on Scenario 1. (D) the *exp/exp* (60%) shedding model on Scenario 2. Fig AM. Contact network G on Day 9. We gathered ten days of HCP movement and interaction data where six days had 14.5-15 hours of observation (Day 2, Day 6, Day 7, Day 8, Day 9, and Day 10). HCP nodes, MWF patient nodes, and TThS patient nodes are depicted in blue, chocolate, and burlywood colors, respectively. We observe a total population of 39 agents (13 HCPs and 26 patients). Fig AN. Cumulative distributions of transmission events over 30 days in Baseline simulation on **Day 9 and** R_0 = 3.0. (A) the *exp/exp* (20%) shedding model on Scenario 1. (B) the *exp/exp* (20%) shedding model on Scenario 2. (C) the *exp/exp* (60%) shedding model on Scenario 1. (D) the *exp/exp* (60%) shedding model on Scenario 2. Fig AO. Attack rates for combinations of inexpensive NPIs on Day 9, $R_0 = 3.0$. (A) the *exp/exp* (20%) shedding model on Scenario 1. (B) the *exp/exp* (20%) shedding model on Scenario 2. (C) the *exp/exp* (60%) shedding model on Scenario 1. (D) the exp/exp (60%) shedding model on Scenario 2. Fig AP. Frequency of replicates as a function of attack rates for combinations of inexpensive NPIs on Day 9, R_0 = 3.0. (A) the exp/exp (20%) shedding model on Scenario 1. (B) the exp/exp (20%) shedding model on Scenario 2. (C) the exp/exp (60%) shedding model on Scenario 1. (D) the exp/exp (60%) shedding model on Scenario 2. Table A. Attack rates for voluntary isolation compliance rate $r_{VI} \in \{0.4, 0.5, 0.6, 0.7, 0.8\}$ under different modeling assumptions. "SM" is short for "shedding models", "B" is short for "Baseline," in Scenario 1 the infection source is a dialysis patient and in Scenario 2 the infection source is a HCP. (PDF)

Acknowledgments

The authors acknowledge the work of Ted Herman in instrumenting the dialysis unit and HCPs in the unit and collecting fine-grained radio signal data. The authors also acknowledge the work of Daniel Sewell and Samuel Justice in processing the raw radio signal data in order to infer time-stamped HCP locations. The authors acknowledge feedback from members of Computational Epidemiology research group at the University of Iowa and the CDC MInD-Healthcare group on an early version of this project.

Author Contributions

Conceptualization: Philip M. Polgreen, Alberto M. Segre, Sriram V. Pemmaraju.

Data curation: Hankyu Jang, Alberto M. Segre.

Formal analysis: Hankyu Jang, Alberto M. Segre, Sriram V. Pemmaraju.

Funding acquisition: Philip M. Polgreen, Alberto M. Segre, Sriram V. Pemmaraju.

Investigation: Hankyu Jang, Philip M. Polgreen, Alberto M. Segre, Sriram V. Pemmaraju.

Methodology: Hankyu Jang, Philip M. Polgreen, Alberto M. Segre, Sriram V. Pemmaraju.

Project administration: Philip M. Polgreen, Alberto M. Segre, Sriram V. Pemmaraju.

Resources: Hankyu Jang, Philip M. Polgreen, Sriram V. Pemmaraju.

Software: Hankyu Jang.

Supervision: Philip M. Polgreen, Alberto M. Segre, Sriram V. Pemmaraju.

Validation: Hankyu Jang, Philip M. Polgreen, Sriram V. Pemmaraju.

Visualization: Hankyu Jang, Sriram V. Pemmaraju.

Writing – original draft: Hankyu Jang, Sriram V. Pemmaraju.

Writing – review & editing: Hankyu Jang, Philip M. Polgreen, Alberto M. Segre, Sriram V. Pemmaraju.

References

- 1. Dong E, Du H, Gardner L. An interactive web-based dashboard to track COVID-19 in real time. The Lancet infectious diseases. 2020. https://doi.org/10.1016/S1473-3099(20)30120-1
- Kambhampati AK, O'Halloran AC, Whitaker M, Magill SS, Chea N, Chai SJ, et al. COVID-19–Associated Hospitalizations Among Health Care Personnel—COVID-NET, 13 States, March 1–May 31, 2020. Morbidity and Mortality Weekly Report. 2020; 69(43):1576. <u>https://doi.org/10.15585/mmwr.mm6943e3</u> PMID: 33119554
- Midturi JK, Ranganath S. Prevention and Treatment of Multidrug-Resistant Organisms in End-Stage Renal Disease. Advances in chronic kidney disease. 2019; 26(1):51–60. https://doi.org/10.1053/j.ackd. 2018.09.003 PMID: 30876618
- Sarnak MJ, Jaber BL. Mortality caused by sepsis in patients with end-stage renal disease compared with the general population. Kidney international. 2000; 58(4):1758–1764. https://doi.org/10.1111/j. 1523-1755.2000.00337.x PMID: 11012910
- Guo H, Liu J, Collins AJ, Foley RN. Pneumonia in incident dialysis patients—the United States renal data system. Nephrology Dialysis Transplantation. 2008; 23(2):680–686. https://doi.org/10.1093/ndt/ gfm474 PMID: 18029368
- Naqvi SB, Collins AJ. Infectious complications in chronic kidney disease. Advances in chronic kidney disease. 2006; 13(3):199–204. https://doi.org/10.1053/j.ackd.2006.04.004 PMID: 16815225
- D'Agata EM, Mount DB, Thayer V, Schaffner W. Hospital-acquired infections among chronic hemodialysis patients. American Journal of Kidney Diseases. 2000; 35(6):1083–1088. https://doi.org/10.1016/ S0272-6386(00)70044-8 PMID: 10845821
- Cohen-Hagai K, Kotliroff A, Rozenberg I, Korzets Z, Zitman-Gal T, Benchetrit S. Effectiveness of Influenza Vaccine in Hemodialyzed Patients: A Retrospective Study. Therapeutic Apheresis and Dialysis. 2019; 23(1):38–43. https://doi.org/10.1111/1744-9987.12742 PMID: 30125465
- Jaber BL. Bacterial infections in hemodialysis patients: pathogenesis and prevention. Kidney international. 2005; 67(6):2508–2519. <u>https://doi.org/10.1111/j.1523-1755.2005.00364.x</u> PMID: 15882306
- Kausz A, Pahari D. The Value of Vaccination in Chronic Kidney Disease. Seminars in Dialysis. 2004; 17 (1):9–11. https://doi.org/10.1111/j.1525-139X.2004.17104.x PMID: 14717802

- Rombolà G, Hedemperger M, Pedrini L, Farina M, Aucella F, Messa P, et al. Practical indications for the prevention and management of SARS-CoV-2 in ambulatory dialysis patients: lessons from the first phase of the epidemics in Lombardy; 2020.
- 12. Goicoechea M, Cámara LAS, Macías N, de Morales AM, Rojas ÁG, Bascuñana A, et al. COVID-19: Clinical course and outcomes of 36 maintenance hemodialysis patients from a single center in Spain. Kidney International. 2020. https://doi.org/10.1016/j.kint.2020.04.031 PMID: 32437770
- Schwierzeck V, König JC, Kühn J, Mellmann A, Correa-Martínez CL, Omran H, et al. First reported nosocomial outbreak of severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2) in a pediatric dialysis unit. Clinical Infectious Diseases. 2020.
- Li J, Xu G. Lessons from the experience in Wuhan to reduce risk of COVID-19 infection in patients undergoing long-term hemodialysis. Clinical Journal of the American Society of Nephrology. 2020; 15 (5):717–719. https://doi.org/10.2215/CJN.03420320 PMID: 32241778
- Meijers B, Messa P, Ronco C. Safeguarding the maintenance hemodialysis patient population during the coronavirus disease 19 pandemic. Blood Purification. 2020; 49(3):259–264. https://doi.org/10.1159/ 000507537 PMID: 32235119
- Mastalerz-Migas A, Gwiazda E, Brydak LB. Effectiveness of influenza vaccine in patients on hemodialysis–a review. Medical science monitor: international medical journal of experimental and clinical research. 2013; 19:1013. https://doi.org/10.12659/MSM.889671
- D'Agata EM. Addressing the problem of multidrug-resistant organisms in dialysis. Clinical Journal of the American Society of Nephrology. 2018; 13(4):666–668. <u>https://doi.org/10.2215/CJN.13781217</u> PMID: 29567862
- Schweiger A, Trevino S, Marschall J. Nosocomial infections in dialysis access. In: Patient Safety in Dialysis Access. vol. 184. Karger Publishers; 2015. p. 205–221.
- Wu C, Chen X, Cai Y, Zhou X, Xu S, Huang H, et al. Risk factors associated with acute respiratory distress syndrome and death in patients with coronavirus disease 2019 pneumonia in Wuhan, China. JAMA internal medicine. 2020. https://doi.org/10.1001/jamainternmed.2020.0994
- 20. Jang H, Justice S, Polgreen PM, Segre AM, Sewell DK, Pemmaraju SV. Evaluating Architectural Changes to Alter Pathogen Dynamics in a Dialysis Unit. In: Proceedings of the 2019 IEEE/ACM International Conference on Advances in Social Networks Analysis and Mining. ASONAM'19. New York, NY, USA: Association for Computing Machinery; 2019. p. 961–968. Available from: https://doi.org/10.1145/ 3341161.3343515.
- Herman T, Pemmaraju SV, Segre AM, Polgreen PM, Curtis DE, Fries JA, et al. Wireless Applications for Hospital Epidemiology. In: 1st Association for Computing Machinery International Workshop on Medical-Grade Wireless Networks. New Orleans, LA; 2009. p. 45–50.
- The National Institute of Diabetes and Digestive and Kidney Diseases Health Information Center. Hemodialysis; 2018. https://www.niddk.nih.gov/health-information/kidney-disease/kidney-failure/ hemodialysis.
- CDC and ASPR. COVID-19 Pandemic Planning Scenarios; 2020. https://www.cdc.gov/coronavirus/ 2019-ncov/hcp/planning-scenarios.html.
- Byambasuren O, Cardona M, Bell K, Clark J, McLaws ML, Glasziou P. Estimating the extent of true asymptomatic COVID-19 and its potential for community transmission: systematic review and metaanalysis. Available at SSRN 3586675. 2020.
- Poletti P, Tirani M, Cereda D, Trentini F, Guzzetta G, Sabatino G, et al. Probability of symptoms and critical disease after SARS-CoV-2 infection. arXiv preprint arXiv:200608471. 2020.
- Oran DP, Topol EJ. Prevalence of Asymptomatic SARS-CoV-2 Infection: A Narrative Review. Annals of Internal Medicine. 2020. https://doi.org/10.7326/M20-3012 PMID: 32491919
- Buitrago-Garcia D, Egli-Gany D, Counotte MJ, Hossmann S, Imeri H, Ipekci AM, et al. Occurrence and transmission potential of asymptomatic and presymptomatic SARS-CoV-2 infections: A living systematic review and meta-analysis. PLoS medicine. 2020; 17(9):e1003346. <u>https://doi.org/10.1371/journal.pmed.1003346</u> PMID: 32960881
- McAloon CG, Collins A, Hunt K, Barber A, Byrne A, Butler F, et al. The incubation period of COVID-19: A rapid systematic review and meta-analysis of observational research. medRxiv. 2020. https://doi.org/ 10.1136/bmjopen-2020-039652 PMID: 32801208
- Mizumoto K, Kagaya K, Zarebski A, Chowell G. Estimating the asymptomatic proportion of coronavirus disease 2019 (COVID-19) cases on board the Diamond Princess cruise ship, Yokohama, Japan, 2020. Eurosurveillance. 2020; 25(10):2000180. https://doi.org/10.2807/1560-7917.ES.2020.25.10.2000180
- Linton NM, Kobayashi T, Yang Y, Hayashi K, Akhmetzhanov AR, Jung Sm, et al. Incubation period and other epidemiological characteristics of 2019 novel coronavirus infections with right truncation: a

statistical analysis of publicly available case data. Journal of clinical medicine. 2020; 9(2):538. https://doi.org/10.3390/jcm9020538 PMID: 32079150

- Li Q, Guan X, Wu P, Wang X, Zhou L, Tong Y, et al. Early transmission dynamics in Wuhan, China, of novel coronavirus–infected pneumonia. New England Journal of Medicine. 2020. https://doi.org/10. 1056/NEJMoa2001316 PMID: 31995857
- Cheng HY, Jian SW, Liu DP, Ng TC, Huang WT, Lin HH. Contact tracing assessment of COVID-19 transmission dynamics in Taiwan and risk at different exposure periods before and after symptom onset. JAMA internal medicine. 2020. https://doi.org/10.1001/jamainternmed.2020.2020 PMID: 32356867
- Bullard J, Dust K, Funk D, Strong JE, Alexander D, Garnett L, et al. Predicting infectious SARS-CoV-2 from diagnostic samples. Clinical Infectious Diseases. 2020. https://doi.org/10.1093/cid/ciaa638 PMID: 32442256
- van Kampen JJ, van de Vijver DA, Fraaij PL, Haagmans BL, Lamers MM, Okba N, et al. Shedding of infectious virus in hospitalized patients with coronavirus disease-2019 (COVID-19): duration and key determinants. medRxiv. 2020.
- Wölfel R, Corman VM, Guggemos W, Seilmaier M, Zange S, Müller MA, et al. Virological assessment of hospitalized patients with COVID-2019. Nature. 2020; 581(7809):465–469. <u>https://doi.org/10.1038/</u> s41586-020-2196-x PMID: 32235945
- He X, Lau EH, Wu P, Deng X, Wang J, Hao X, et al. Temporal dynamics in viral shedding and transmissibility of COVID-19. Nature medicine. 2020; 26(5):672–675. <u>https://doi.org/10.1038/s41591-020-0869-</u> 5 PMID: 32296168
- Carrat F, Vergu E, Ferguson NM, Lemaitre M, Cauchemez S, Leach S, et al. Time lines of infection and disease in human influenza: a review of volunteer challenge studies. American journal of epidemiology. 2008; 167(7):775–785. https://doi.org/10.1093/aje/kwm375 PMID: 18230677
- Zou L, Ruan F, Huang M, Liang L, Huang H, Hong Z, et al. SARS-CoV-2 viral load in upper respiratory specimens of infected patients. New England Journal of Medicine. 2020; 382(12):1177–1179. <u>https:// doi.org/10.1056/NEJMc2001737 PMID: 32074444</u>
- Casey M, Griffin J, McAloon CG, Byrne AW, Madden JM, McEvoy D, et al. Estimating pre-symptomatic transmission of COVID-19: a secondary analysis using published data. medRxiv. 2020.
- Zhou R, Li F, Chen F, Liu H, Zheng J, Lei C, et al. Viral dynamics in asymptomatic patients with COVID-19. International Journal of Infectious Diseases. 2020. https://doi.org/10.1016/j.ijid.2020.05.030 PMID: 32437933
- Noh JY, Yoon JG, Seong H, Choi WS, Sohn JW, Cheong HJ, et al. Asymptomatic infection and atypical manifestations of COVID-19: comparison of viral shedding duration. The Journal of Infection. 2020. https://doi.org/10.1016/j.jinf.2020.05.035 PMID: 32445728
- **42.** Wu JT, Leung K, Leung GM. Nowcasting and forecasting the potential domestic and international spread of the 2019-nCoV outbreak originating in Wuhan, China: a modelling study. The Lancet. 2020; 395(10225):689–697. https://doi.org/10.1016/S0140-6736(20)30260-9 PMID: 32014114
- Riou J, Althaus CL. Pattern of early human-to-human transmission of Wuhan 2019 novel coronavirus (2019-nCoV), December 2019 to January 2020. Eurosurveillance. 2020; 25(4):2000058. https://doi.org/ 10.2807/1560-7917.ES.2020.25.4.2000058
- 44. Microsoft. Bing COVID-19 Tracker; 2020. Available from: http://www.bing.com/covid.
- Bałazy A, Toivola M, Adhikari A, Sivasubramani SK, Reponen T, Grinshpun SA. Do N95 respirators provide 95% protection level against airborne viruses, and how adequate are surgical masks? American journal of infection control. 2006; 34(2):51–57. <u>https://doi.org/10.1016/j.ajic.2005.08.018</u> PMID: 16490606
- 46. Bischoff WE, Reid T, Russell GB, Peters TR. Transocular entry of seasonal influenza–attenuated virus aerosols and the efficacy of N95 respirators, surgical masks, and eye protection in humans. Journal of Infectious Diseases. 2011; 204(2):193–199. https://doi.org/10.1093/infdis/jir238 PMID: 21673029
- Centers for Disease Control and Prevention (CDC). Social Distancing, Quarantine, and Isolation: Keep Your Distance to Slow the Spread; 2020. https://www.cdc.gov/coronavirus/2019-ncov/prevent-gettingsick/social-distancing.html.
- Jang H. COVID-19 Modeling and Non-Pharmaceutical Interventions in a Dialysis Unit; GitHub repository; 2020. https://github.com/HankyuJang/Dialysis_COVID19.
- 49. Park S, Kim Y, Yi S, Lee S, Na B, Kim C, et al. Coronavirus Disease Outbreak in Call Center, South Korea. Emerging Infectious Diseases. 2020; 26(8). https://doi.org/10.3201/eid2608.201274 PMID: 32324530
- Centers for Disease Control and Prevention (CDC). Strategies to Mitigate Healthcare Personnel Staffing Shortages; 2020. https://www.cdc.gov/coronavirus/2019-ncov/hcp/mitigating-staff-shortages.html.

- Schwartz A, Stiegel M, Greeson N, Vogel A, Thomann W, Brown M, et al. Decontamination and reuse of N95 respirators with hydrogen peroxide vapor to address worldwide personal protective equipment shortages during the SARS-CoV-2 (COVID-19) pandemic; 2020.
- 52. Ma Y, Diao B, Lv X, Zhu J, Liang W, Liu L, et al. COVID-19 in hemodialysis (HD) patients: Report from one HD center in Wuhan, China. medRxiv. 2020.
- Yau K, Muller MP, Lin M, Siddiqui N, Neskovic S, Shokar G, et al. COVID-19 outbreak in an urban hemodialysis unit. American Journal of Kidney Diseases. 2020; 76(5):690–695. https://doi.org/10.1053/ j.ajkd.2020.07.001 PMID: 32681983
- Ahmed AE, Alshukairi AN, Al-Jahdali H, Alaqeel M, Siddiq SS, Alsaab HA, et al. Development of a riskprediction model for Middle East respiratory syndrome coronavirus infection in dialysis patients. Hemodialysis International. 2018; 22(4):474–479. https://doi.org/10.1111/hdi.12661 PMID: 29656480
- 55. Hastings DL, Tokars JI, Aziz IZAA, Alkhaldi KZ, Bensadek AT, Alraddadi BM, et al. Outbreak of Middle East respiratory syndrome at tertiary care hospital, Jeddah, Saudi Arabia, 2014. Emerging infectious diseases. 2016; 22(5):794. https://doi.org/10.3201/eid2205.151797 PMID: 27089550
- D'Agata EM, Green WK, Schulman G, Li H, Tang YW, Schaffner W. Vancomycin-resistant enterococci among chronic hemodialysis patients: a prospective study of acquisition. Clinical infectious diseases. 2001; 32(1):23–29. https://doi.org/10.1086/317549 PMID: 11112676
- Pop-Vicas A, Strom J, Stanley K, D'Agata EM. Multidrug-resistant gram-negative bacteria among patients who require chronic hemodialysis. Clinical Journal of the American Society of Nephrology. 2008; 3(3):752–758. https://doi.org/10.2215/CJN.04651107 PMID: 18322047
- Snyder GM, D'Agata EM. Novel antimicrobial-resistant bacteria among patients requiring chronic hemodialysis. Current opinion in nephrology and hypertension. 2012; 21(2):211–215. https://doi.org/10.1097/ MNH.0b013e328350089b PMID: 22240441
- 59. Wares JR, Lawson B, Shemin D, D'Agata EM. Evaluating infection prevention strategies in out-patient dialysis units using agent-based modeling. Plos one. 2016; 11(5):e0153820. <u>https://doi.org/10.1371/journal.pone.0153820 PMID: 27195984</u>
- **60.** Ajmal F. Association of Freestanding Dialysis Facility Size, Quality Incentive Program Scores and Patient Survival. Doctoral dissertation. 2017.
- 61. General Information | MRSA | CDC;. Available from: https://www.cdc.gov/mrsa/community/index.html.
- CDC. Community, Work, and School;. Available from: <u>https://www.cdc.gov/coronavirus/2019-ncov/</u> community/parks-rec/public-beaches.html.
- van Doremalen N, Bushmaker T, Morris DH, Holbrook MG, Gamble A, Williamson BN, et al. Aerosol and surface stability of SARS-CoV-2 as compared with SARS-CoV-1. New England Journal of Medicine. 2020; 382(16):1564–1567. https://doi.org/10.1056/NEJMc2004973
- Rincón A, Moreso F, López-Herradón A, Fernández-Robres MA, Cidraque I, Nin J, et al. The keys to control a COVID-19 outbreak in a haemodialysis unit. Clinical kidney journal. 2020; 13(4):542–549. https://doi.org/10.1093/ckj/sfaa119 PMID: 32885797